Author’s response to reviews

Title: Factors affecting early identification of pregnant women by community health workers in Morogoro, Tanzania

Authors:

GASTO FRUMENCE (frumencegasto@yahoo.co.uk)
M Goodman (maddgoodm@gmail.com)
J Chebet (joychebbet@gmail.com)
I Mosha (ihmosha@yahoo.co.uk)
D Bishanga (Dunstan.Bishanga@jhpiego.org)
D Chitama (dereck.chitama@gmail.com)
PJ Winch (pwinch@jhu.edu)
J Killewo (jkillewo@yahoo.co.uk)
AH Baqui (abaqui@jhu.edu)

Version: 3 Date: 08 Jun 2019

Author’s response to reviews:

Chief Editor,

BMC Public Health

RE: SUBMISSION OF MANUSCRIPT TITLED ‘FACTORS AFFECTING EARLY IDENTIFICATION OF PREGNANT WOMEN BY COMMUNITY HEALTH WORKERS IN MOROGORO, TANZANIA’ FOR PUBLICATION CONSIDERATION IN YOUR JOURNAL.

Thank you very much for informing us that potentially our manuscript has been accepted for publication in your journal after addressing minor comments.
Kindly be informed that we have addressed minor comments from the editor and one reviewer.

One reviewer recommended that we should not start a sentence with abbreviation. We have corrected this error for the entire document.

The editor requested us to state if the focus group discussion guides were developed for this study or they were adopted from previous studies. We have stated that the guides were developed for this specific study and we have attached them as supplementary files as requested.

One of the authors (Peter Winch) who is a native English speaker has edited the manuscript and corrected any grammatical errors as recommended.

It is our hope that our manuscript will now be published in your journal

I am looking forward to your positive response.

Yours truly,

Gasto Frumence (corresponding author)

Senior Lecturer in Development Studies